Author's response to reviews

Title: Determinants of Maternal Health Care Utilization in Holeta Town, Central Ethiopia

Authors:

Kidist B Adinew (kidistbirmeta@gmail.com)
Desalegn W Kifle (desalegn01@gmail.com)

Version: 2 Date: 31 October 2012

Author's response to reviews: see over
Determinants of Maternal Health Care Utilization in Holeta Town, Central Ethiopia

Kidist Birmeta*¹, Desalegn Woldeyohannes², Yohannes Dibaba³

¹ College of Development Studies, Center for Population Studies, Addis Ababa University, Addis Ababa, Ethiopia
² Department of Public Health, School of Medical and Health Sciences, Addis Ababa Science and Technology University, Addis Ababa, Ethiopia
³ Department of population and family health, Jimma University, Jimma, Ethiopia

*Corresponding author

Email addresses:
KB: kidistbirmeta@gmail.com
DW: desalegn01@gmail.com
YD: yohannesdw@yahoo.com